

ANSWERING REVIEWERS

November 21, 2014



Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 13234/Invitation ID 00738190 - review.doc).

Title: Melkersson-Rosenthal syndrome.

Author: Ioannis Chatzistefanou, Petros Zikos, Grigorios Venetis, Konstantinos Vahtsevanos, Konstantinos Antoniadis.

Name of Journal: *World Journal of Stomatology*

ESPS Manuscript NO: 13234/Invitation ID 00738190

The manuscript has been improved according to the suggestions of reviewers:

- 1 Format has been updated
- 2 Revision has been made according to the suggestions of the reviewer
 - (1) The manuscript has been revised regarding overlapping text.
 - (2) All the references has been properly cited.
- 3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Stomatology*.

Sincerely yours,

A handwritten signature in blue ink, appearing to be 'I. Chatzistefanou', written over a horizontal line.

Ioannis CHATZISTEFANOU, MD, DDS
Dept. of Oral and Maxillofacial Surgery
Aristotle University of Thessaloniki
"G. Papanikolaou" General Hospital, Exohi 57010,
Thessaloniki, Greece
Tel: +30 2310 760124
Fax: +30 2310 760124
E-mail: ioannis.chatzistefanou@gmail.com